

Edinburgh Research Explorer

What can cohort studies in the dog tell us?

Citation for published version:

Pugh, C, Bronsvoort, M, Handel, IG, Summers, KM & Clements, DN 2014, 'What can cohort studies in the dog tell us?' Canine Genetics and Epidemiology, vol. 1, no. 1, pp. 5. DOI: 10.1186/2052-6687-1-5

Digital Object Identifier (DOI):

10.1186/2052-6687-1-5

Link:

Link to publication record in Edinburgh Research Explorer

Document Version:

Publisher's PDF, also known as Version of record

Published In:

Canine Genetics and Epidemiology

Publisher Rights Statement:

© 2014 Pugh et al.; licensee BioMed Central Ltd.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/2.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly credited. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

General rights

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy

The University of Edinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.





REVIEW Open Access

What can cohort studies in the dog tell us?

Carys A Pugh*, Barend M de C Bronsvoort, Ian G Handel, Kim M Summers and Dylan N Clements

Abstract

This paper addresses the use of cohort studies in canine medicine to date and highlights the benefits of wider use of such studies in the future. Uniquely amongst observational studies, cohort studies offer the investigator an opportunity to assess the temporal relationship between hypothesised risk factors and diseases. In human medicine cohort studies were initially used to investigate specific exposures but there has been a movement in recent years to more broadly assess the impact of complex lifestyles on morbidity and mortality. Such studies do not focus on narrow prior hypotheses but rather generate new theories about the impact of environmental and genetic risk factors on disease. Unfortunately cohort studies are expensive both in terms of initial investment and on-going costs. There is inevitably a delay between set up and the reporting of meaningful results. Expense and time constraints are likely why this study design has been used sparingly in the field of canine health studies. Despite their rather limited numbers, canine cohort studies have made a valuable contribution to the understanding of dog health, in areas such as the dynamics of infectious disease. Individual exposures such as neutering and dietary restriction have also been directly investigated. More recently, following the trend in human health, large cohort studies have been set up to assess the wider impact of dog lifestyle on their health. Such studies have the potential to develop and test hypotheses and stimulate new theories regarding the maintenance of life-long health in canine populations.

Keywords: Canine, Cohort, Longitudinal, Epidemiology

Lay summary

Cohort studies involve the repeated collection of information through time. Investigators are able to assess whether exposure to a particular risk factor (these could be environmental, diet, lifestyle, genetic etc) is associated with, and followed by, clinical outcomes in individuals. This paper highlights the benefits of wider use of such studies in the future. Thus researchers collect information (e.g. clinical data and records), rather than DNA samples to use in genetic studies. Information is longitudinally collected at regular time intervals, and allows investigators to assess whether exposure to particular risk factors (these could be environmental, diet, lifestyle, genetic, etc) are associated with, and followed by, clinical outcomes in individuals. This paper highlights the benefits of wider use of such studies in the future.

Uniquely amongst observational studies, cohort studies offer researchers an opportunity to assess the time relationship between suggested risk factors and diseases. In human medicine, cohort studies were originally used to investigate

specific exposures but there has been a recent move to more broadly assessing the impact of complex lifestyles on life and death. Such studies can generate new theories about the impact of environmental and genetic risk factors on disease. However, cohort studies are expensive and there is, inevitably, a delay between setting them up and the final reporting of meaningful results. Expense and time constraints are why this type of study has been used sparingly in canine health studies. Despite limited numbers, canine cohort studies have made a valuable contribution to the understanding of dog health, in areas such as infectious disease. Individual exposures such as neutering and dietary restriction have also been investigated. More recently, large cohort studies have been set up to assess the impact of dog lifestyle on their health. Such studies have the potential to develop and test hypotheses (ideas) and stimulate new theories regarding the maintenance of life-long health in canine populations.

Introduction

Understanding the factors relating to disease in a population is important for anticipating and dealing with health care needs. The health of populations can be

^{*} Correspondence: Carys.Pugh@roslin.ed.ac.uk The Roslin Institute and The Royal (Dick) School of Veterinary Studies, The University of Edinburgh, Easter Bush, Midlothian EH25 9RG, UK



studied in a number of ways. Beyond descriptive approaches, analytical studies can be split into experimental and observational investigations. Dohoo *et al.* [1] distinguished observational studies from experimental studies, where investigators control the allocation of subjects to study groups, by suggesting that in observational studies, investigators "try not to influence the natural course of events for the study subjects".

Epidemiologists traditional divide observational studies into case-control, cross-sectional or cohort study designs [1,2]. The advantages and disadvantages of each of these study types, particularly with regard to susceptibility to bias, are fully described in Table 1. In brief however, case-control studies are particularly useful for rare diseases but lack an ability to clarify temporal relationships between events and exposures. Cross-sectional studies can be performed at a single time point and allow investigators to seek associations between potential risk factors and outcomes, but again do not allow the assessment of temporal dependencies. Cohort studies, where individuals are tracked through time, solve this problem as investigators can assess whether risk factor exposures are *followed* by outcomes in individuals. This element of time dependency is crucial to infer causation between risk factors and disease, and to understand transmission dynamics of infectious diseases. Further, cohort studies lend themselves to analysis of the effect of long-term exposure to a risk factor or treatment and, with targeted recruitment, are ideally suited to examine the effect of rare risk factors. Unfortunately cohort studies necessarily involve a large investment of time and finances, both to set up and maintain, and have therefore been used sparingly in the field of canine disease. In this review we will discuss studies found in broader medical literature before describing the types of canine cohort studies reported to date. The techniques found in human medicine may be applied to canine epidemiology with immense potential for health advances.

Review

The benefits of cohort studies: comparative examples

One of the most widely renowned cohort studies in human medicine is the Framingham Heart Study. Researchers recruited a group of over 5,000 women and men aged between 30 and 62 years old living in Framingham, Massachusetts in 1948. The cohort were evaluated every two years regarding their medical status and lifestyles, including physical examinations and collection of biological samples for laboratory testing. The study identified many of the major cardiovascular disease risk factors which we

Table 1 The advantages and disadvantages of different observational study types

Study type	Potential goals	Advantages	Disadvantages
Cross-sectional	- Population prevalence of exposure and/ or outcome	- Relatively simple	- Poor for rarer exposures and outcomes
	- Associations between exposures and outcomes	- Relatively cheap	 No causality may be inferred as exposures and outcomes are measured contemporaneously
		- Relatively quick	- Highly susceptible to information bias
		- Good for common conditions and exposures	
		- May assess multiple exposures and outcomes	
		- Good for initial assessment of an exposure or outcome	
Case-control	- Associations between exposures and outcome	- Relatively cheap	- Choice of controls notoriously difficult
	ratio between exposure(s) in controls and	- Relatively quick	- May only examine one outcome
		- May assess long latent periods	- Odds ratio not an intuitive measure
		- Good for rarer outcomes	 Highly susceptible to selection and information bias and population stratification
Cohort	- Incidence rates	- Good for rare exposures	- Not simple
	- Temporal associations between exposures and outcomes	- May examine multiple exposures and outcomes	- Not cheap
		- May assess long latent periods	- Not quick (unless retrospective)
		- May assess temporal relationship between exposure and outcome inferring causality	- Highly susceptible to retention bias
			- Susceptible to sampling bias

take for granted today, such as high blood pressure, high blood cholesterol, smoking, obesity and diabetes [3]. The analysis of the Framingham cohort has resulted in over 2,000 peer reviewed publications, and aptly demonstrates how the detailed, repeated evaluation of modestly sized cohort groups can result in the identification of risk factors for disease which have global significance.

Another early cohort study of human health was undertaken in the UK in 1951. The aim was to address concerns about an observed association between smoking and lung cancer. To examine the question of causality, the study was designed to determine whether it was possible to predict someone's risk of developing lung cancer from their smoking habits earlier in life [4]. Over 40,000 doctors were recruited, which was over two thirds of the doctors on the British Medical Register at the time. The study went on to investigate the impact of smoking on diseases beyond lung cancer, including vascular disease and other neoplasias [5]. Ultimately the cohort was so valuable that the members were followed for their lifetime and the last questionnaire was sent out some 50 years later.

Two more recent studies which have illustrated the power of large scale cohort studies are the Avon Longitudinal Study of Parents and Children (ALSPAC) [6] and the Italian NINFEA cohort [7]. Both are birth cohorts, initially designed without specific hypotheses in mind. Instead they set out to collect information on a variety of exposures to broadly investigate pregnancy and the early life of children. In the case of ALSPAC, investigations went back even earlier, with assessment of antenatal risks, such as the impact of maternal drinking prior to conception and in early pregnancy on birth weight [8].

The ALSPAC study team faced great difficulty obtaining funding in the initial years of the project [9]. As time passed and significant risk factors started to be found and reported, it became more widely recognised that the cohort was an incredible resource that should be maintained in the long-term. This open-ended investigative approach resulted in the identification of a range of phenotypes and influencing factors that could not have been predicted by the investigators at the start of the study. The costs of recruiting the cohort would have been wasted if contact with members were lost before these discoveries could be made.

Analyses of the ALSPAC cohort did not stop with exploration of early-life influences. As the costs of collecting, archiving and analysing DNA reduced it became possible to add genetic data to the wealth of phenotypic data and explore the interaction of genotype with other variables. Over more than 20 years the ALSPAC team moved from having a relationship with pregnant mothers to having a relationship with the children from those pregnancies. These children have grown to start their own families and the next

generation are also being recruited into the study. A wealth of discoveries guiding national public health policy have been made during the study. These include understanding the influence of sleeping position on the risk of cot death [10,11] and the benefits of eating oily fish on children's mental development [12], both of which have directly led to the development of guidelines for best practice.

Between 1996 and 2001, the Million Women Study recruited women over 50 in the UK [13]. Recruitment through breast cancer screening centres built-in a reliable method of ascertaining the primary outcome of interest – the incidence of breast cancer. Environmental influences were captured in a lifestyle questionnaire that was completed at recruitment and periodically thereafter. Information regarding other disease events such as incidence of fractures was also collected via the follow-up questionnaires [14].

The main finding from the Million Women Study regarding the impact of Hormone Replacement Therapy (HRT) on the incidence of breast cancer [15] remains controversial. An increased incidence of breast cancer was found in the women taking HRT but it has subsequently been argued that these women were more likely to be tested for breast cancer, resulting in increased diagnoses. The women involved were not randomly assigned to receive HRT so the potential for confounding cannot be ignored. Nevertheless, the study built on results from earlier cross-sectional studies and it had enormous power to detect associations. As the women were followed with time, causal inference is possible. At the very least the results of the many publications about the cohort will influence the direction of future randomised controlled trials to try and definitively determine causal relationships.

The value of cohorts has been recognised and data collected previously are increasingly the foundation for further analysis. For example, a team from Edinburgh University took advantage of historic data collection to develop a cohort of people with results that span over 80 years. In Scotland 95% of children born in 1921 and 1936 were given an intelligence test at the age of 11. The team recruited a subset of the survivors from these tested cohorts some 60-70 years later to investigate their cognitive function [16] and the environmental and genetic influences upon them. Their continuing assessment of cognitive function has led to the discovery of an association between carrying the APOE E4 allele (also associated with Alzheimer's disease) and non-pathological cognitive decline [17]. The cohort is a unique resource for the investigation of the effects of aging on cognition and it continues as participants enter their tenth decade.

While the benefits of cohort studies are well understood (Table 1), the extended time to finding results and relatively high costs are undeniable. In part to address these costs, the US Department of Defense started to

move cohort studies into the internet age when they set up the Millennium Cohort Study of US military personnel [18]. Current and ex-military personnel were recruited and offered the chance to answer the questionnaire by post or online. The financial savings associated with participants replying online were such that they offered a \$5 incentive and still estimated their savings per online response at \$50 compared to those responding by post [19]. As internet access has increased, epidemiological studies have gradually made greater use of the technology. The NINFEA cohort is based entirely online [20]. Whilst the costs of setting up and maintaining functional and appealing web portals are not insignificant, studies are now possible that would not have been feasible if based on face-to-face, telephone or postal questionnaires. Building on this experience of human studies, canine cohort studies that would have been inconceivable are now financially viable and the potential to exploit this avenue of research is immense.

Canine cohort studies

Despite the extensive number of findings uncovered by human cohort studies, the design has not been widely used in canine research in the past. As discussed, the cost and time burdens can be prohibitively high. A number of canine cohort studies have been reported and in each case attempts have been made to overcome the associated financial burden. The different strategies used are discussed below and their merits summarised in Table 2.

Retrospective methods

Retrospective cohort studies involve looking back at individuals after the events of interest have occurred (for example disease incidence, death or pregnancy) and the follow-up period has ended [21]. These studies can be undertaken on a large scale with relatively little lead time or up-front costs by using pre-existing databases such as those maintained by insurance companies and groups of secondary veterinary hospitals or primary clinics.

Insurance databases in particular are an extremely valuable resource and are discussed in detail by O'Neill et al. [22]. There is a long tradition of pet insurance in Sweden and Agria insure approximately 40% of Swedish dogs [23]. Their database provides a powerful measure of events in the Swedish pet canine population [24]. Such large electronic resources offer the chance to study incidence rates and survival time from diagnosis for specific diseases, such as mammary tumours [25]. However there is no requirement for private companies to make their data available. When using insurance databases, there is likely to be bias relating to the non-random socioeconomic status of owners who insure their pets and

to specific insurance policy exclusions such as preexisting conditions and age limits. In addition, in countries where dog insurance rates are low, the resource would be even less representative of the population as a whole.

Veterinary medical databases provide an alternative resource of information on the health of populations [22]. They have the advantage of that they can be linked to ancillary resources (such as radiographic archives and biological samples). However, the plethora of recording systems, and lack of agreement of diagnostic criteria for the definition of specific diseases, makes them cumbersome to use and extracting and extrapolating data is difficult. With modern textural mining tools there is scope to revisit this area [22] but the challenge of collating records from diverse recording systems remains. Further, when these databases rely solely on groups of specialist hospitals, there is the risk of referral bias as demonstrated by Bartlett *et al.* [26].

Risk factor studies using both insurance and veterinary medical databases are also limited by the type of data collected. In both cases, the data refer to phenotype of the dog but not their wider environment. Postcode (location) data have been used to assess the spatial distribution of atopic dermatitis [27] but the impact of the dogs' lifestyles is not available from such records. For example Glickman et al. [28] were able to investigate a link between severity of periodontal disease in dogs and subsequent chronic azotemic kidney disease (kidney disease causing high levels of blood urea and creatinine) because both diseases were recorded in clinical records, but environmental risk factors like diet could not be considered. This is a major limitation of such databases; otherwise their data on multiple disease outcomes, covering large numbers of dogs from different breeds, would be unparalleled in terms of potential for use in investigations.

Prospective methods

Prospective studies are set up before the outcome of interest occurs and allow investigators to pre-select study subjects and specifically determine which data they wish to collect [21].

Prospective methods: time-limited

Limiting the time at risk has been used to minimise the costs of studies where pre-existing data are not available. This also helps reduce bias through loss to follow up. A wealth of investigations have utilised this methodology, such as those investigating the spread of Leishmaniasis and other vector borne diseases in dog cohorts. Studies investigating disease incidence [29-31], detection methods [32-36] and the impact of a culling regime [37] have all used this approach. Cohort methodology was necessary in each study

Study type	Data source(s)	Advantages	Disadvantages
Retrospective	Pre-existing insurance databases	- Relatively cheap	- Non-standardised diagnostic criteria
		- Relatively quick	- Poor generalisability in countries with high uninsured population
		 May assess multiple clinical exposures and outcomes 	- No requirement for insurance data to be made available
		- May assess long latent periods	
		- Recruitment and retention simple	
	Pre-existing databases from secondary veterinary hospitals	- Relatively cheap	- Non-standardised diagnostic criteria
		- Relatively quick	- Non-standardised recording systems
		- May assess multiple clinical exposures and outcomes	- No knowledge of wider environmental exposures
		- Potential to use ancillary resources	- Potential for referral and geographical bias
		- May assess long latent periods	
		- Good for examining serious illnesses	
		- Recruitment and retention simple	
	Pre-existing databases from primary veterinary clinics	- Relatively cheap	- Non-standardised diagnostic criteria
		- Relatively quick	- Non-standardised recording systems
		- May assess multiple clinical exposures and outcomes	- No knowledge of wider environmental exposures
		- Recruitment simple	- Potential for retention bias as owners move practices
Prospective: Time Limited	According to study protocol: May include investigators, veterinarians, breeders and owners	- Costs and time limited according to length of the study	 Necessarily time limited so unable to assess long-term exp sures and long latent periods
		 May assess multiple exposures and outcomes including wider environmental exposures 	- Recruitment not simple
		- Good for the study of infectious diseases	
		- Diagnostic criteria set according to study protocol	
		- Retention bias is minimised	
Prospective: Single issue	According to study protocol: May include investigators, veterinarians, breeders and owners	- Potential to examine a single issue in great detail	- Not quick
		- Diagnostic criteria set according to study protocol	- Potentially very expensive
		- May assess wider environmental exposures	- Recruitment not simple
			- Potential for retention bias in uncontrolled conditions
			- May only examine multiple exposures <i>OR</i> multiple outcom

Table 2 The advantages and disadvantages of different cohort study types (Continued)

Prospective: According to study protocol: Animals typically population- May assess multiple exposures - Not quick Hypothesis based but data maybe generated by investigators, veterinarand outcomes including wider ians, breeders and owners generation environmental exposures - Diagnostic criteria set - Not cheap according to study protocol - Potential to describe health - Delay to results and lack of and lifestyle of current specific focus make funding difficult population - Potential to assess the broad - Recruitment not simple impact of lifestyle on disease - Potential to generate new - High susceptibility to hypotheses retention bias - Potential for poor diagnostic accuracy if reliant on ownerreporting

but cost minimisation and swift reporting were facilitated by limiting the follow-up times to from one to three vector seasons.

Prospective methods: single factor

If time is not constrained, then the focus or numbers of dogs in a study may be narrowed. Perhaps the best example of this comes from a study of dietary restriction using a small group of Labrador Retrievers (48 dogs) in an experimental setting. This controlled trial has yielded an array of findings on the effect of dietary restriction on mortality [38,39], immune function [40], and developmental joint disease [41-49]. The time-span and depth of this trial (including blood sampling and radiography at regular intervals) made it prohibitively expensive to perform on a larger scale but data on specific aspects, such as the life-long progression of osteoarthritis, could only be collected by following a cohort longitudinally in this manner.

Dobson *et al.* [50] undertook a study with a similarly narrow focus but were able to recruit dogs from the normal pet population in the UK. Following 174 flat-coated retrievers for up to 10 years they investigated the impact of neoplasia on mortality in that breed. Costs were also minimised in this case by contacting recruited owners just once per year for a health update and asking them to proactively contact the investigators if their dog fell ill. The study demonstrated that over 40% of the dogs died as a result of neoplasia, reducing their lifespan by three years compared to those that died from other causes.

Recruiting a large enough cohort to give the required power for an investigation and retaining that cohort to minimise bias are both key to the success of population-based cohort studies. Thrusfield *et al.* [51] studied a cohort of bitches for up to five years in an attempt to assess the impact of neutering on urinary incontinence. The onus for recruiting and maintaining the cohort was

placed on volunteering veterinary surgeons. Perhaps because of this responsibility, some difficulty was encountered recruiting veterinarians to participate; whilst 233 initially agreed, only 16 went on to return data (a 7% response rate). The authors made every effort to minimise bias through randomisation techniques but the potential impact of selection bias on the study should not be overlooked.

Each veterinarian was asked to recruit 40 female puppies from their practices. Should these bitches subsequently become incontinent then they were no longer followed, whilst, by design, the remaining (continent) cohort were to be followed for five years. The veterinarians received letters encouraging them to continue with the study at one and three years, and a request to contact the involved owners to check that their dogs were not incontinent after five years. The authors cite slow initial recruitment as the main reason why only 504 dogs from an original 809 were followed for the full five years. They do not directly address how many of the remaining 305 dogs were lost to follow-up (only 22 developed incontinence), but the potential impact of retention bias on their results cannot be ignored. Nevertheless, by focussing on a single phenotype and spreading the responsibility for dealing with recruited animals amongst a number of veterinarians, it was possible to follow enough dogs to determine that neutered bitches had a risk of urinary incontinence that was nearly eight-fold that of intact bitches.

Prospective methods: hypothesis generation

Beyond studies that focus on one disease or one exposure, there has been a movement in canine epidemiology toward the broader studies undertaken in human medicine such as the example of ALSPAC mentioned above [9]. These studies do not necessarily aim to test a single hypothesis but rather gather data to identify new areas of investigation. In canine medicine, questionnaires have

been developed that cover a wide range of potential exposures and disease outcomes and they are directed at breeders, owners and veterinarians. These studies have the disadvantage of relying on non-standardised data inputs where each animal is assessed by a different person with disparate (or no) training. However the studies are able to recruit more participants, and their subjects are more representative of dog lifestyle in the wider population than those followed under controlled conditions.

A 10-year cohort study of pedigree Boxers in the Netherlands recruited over 90% of the litters born in 14 months of 1994–5, initially comprising 2629 puppies. The study used diary-format records and face-to-face assessment with the breeders but moved on to sixmonthly questionnaires with owners. Pre-weaning mortality [52,53] and post-weaning mortality [54] were assessed and, due to the large numbers of dogs involved in the study, all with pedigree information, the investigators were able to make heritability estimates for phenotypes [55] and common diseases such as cryptorchidism (failure of one or both testes to descend to the scrotum), cranial cruciate disease (degeneration of the cranial cruciate ligament) and epilepsy [56] (a neurological disease characterised by the development of seizures) and hip dysplasia (a developmental malformation of the hip joints) [57].

Similarly a group in Norway followed a cohort of 700 dogs from four large breeds. Again they gave questionnaires to breeders and owners but they also involved the dogs' veterinarians. To date they have published studies on the prevalence and risk factors of neonatal mortality [58], the effect of weight and growth rates on the development of hip dysplasia [59] and the incidence and risk factors associated with vomiting and diarrhoea [60].

Relatively newly created is the Dogslife Project [61], which is focussed on the owners of Kennel Club registered Labrador Retrievers in the UK [62]. It is limiting costs by utilising a website-based questionnaire and has recruited over 4,200 dogs in three and a half years. As a prospective study, it was possible to specifically tailor the questionnaire to address areas of interest. Data collection includes detail regarding phenotype and lifestyle which will be examined with reference to dog health. Like the studies in Norway and the Netherlands, the Dogslife Project is an attempt to develop a large-scale cohort of dogs with thoroughly documented history, similar to those cohorts found in human medicine.

The future of canine cohort research

With the relative dearth of cohort studies in canines to date, there is scope to address new questions in the future. For example, the cohort of Dutch Boxer dogs discussed earlier were reported to have a pre-weaning mortality rate over 20% [52]. Such a loss is a clear welfare problem for

dogs and more detailed studies of potential risk factors could have a great impact. Indrebø et al. [58], Nielen et al. [52] and van der Beek et al. [53] each address early mortality through cohort studies but their findings focus largely on factors from birth onward. Van der Beek et al. [53] included an analysis of inbreeding coefficients but found that genetic effects in general had less effect than environmental effects at puppy and litter level. Relatively short cohort studies including the lifestyle of the dam prior to birth may shed new light on risk factors associated with both still-births and early mortality, minimising distress in owners who are currently unable to prevent early losses.

Cohort studies such as those undertaken in human medicine could play a vital role if veterinarians are to be able to offer advice to owners on minimising the risks of developing disease and injury. Beyond death in very early life, morbidity and mortality in dogs in developed nations reflects the epidemiological shift in morbidity and mortality in human medicine from infectious diseases to noncommunicable diseases. This shift is increasingly relevant in canine health as vaccination, antibiotics and better veterinary care ensure that more dogs in developed nations live to suffer from developmental diseases and diseases of aging. Bonnett et al. [63] demonstrated that whilst the highest mortality rate in dogs over six weeks of age in Sweden was trauma (typically car accidents), the next highest rate was due to tumours, followed by locomotor problems. Cohort studies of canine lifestyle have the power to investigate the risk factors associated with developing these noncommunicable diseases, facilitated by the release of a draft canine genome sequence [64,65] and the increasing access to high density genotyping and eventually low cost whole genome sequencing. Since the dog has a shorter lifespan than humans, associations between genetic variation and disease that are also relevant to human aging are likely to be revealed.

Human medicine is again ahead of the veterinary field with regard to incorporating biological data in cohort studies. UK Biobank that have recruited 500,000 people between 40-69 years of age. The investigative team collect blood, saliva and urine samples, phenotypic data and the agreement of all participants to have their health status followed. The collection of genetic information in particular adds a new element to the traditional cohort study, and with such a large cohort the potential power to detect risk factors involving genetic-environmental interactions is enormous. Projects on such a scale are currently financially prohibitive in dogs, but projects on a smaller scale such as Dogslife [61] have collected buccal swabs for DNA extraction from a subset of their cohort enabling comparisons of genotype with phenotype. Should such sampling be repeated throughout the lives of the dogs, investigations of genetic and epigenetic changes throughout that lifetime and comparison with concurrent phenotypic data would potentially permit investigators to seek environmental factors associated with genetic variation, disease and aging. The merging of lifestyle and whole genome data should increasingly reveal associations between genotype and environment in the dog and ultimately in humans as well.

Conclusions

Cohort studies have already yielded results in the field of canine health. With the advent of large databases and internet technology the costs of such studies are being reduced to the point whereby large-scale studies are possible in canine populations. The potential to identify risk factors and inform an evidenced-based medicine approach to preventative health measures in dogs mean that cohort studies can have a great impact on dog health and welfare. Given how long it takes to achieve results from prospective studies, the time to start is now.

Abbreviations

ALSPAC: Avon longitudinal study of parents and children; HRT: Hormone replacement therapy.

Competing interests

The authors of this review are involved in the development of the Dogslife project [62]. They derive no financial benefit from their involvement.

Authors' contribution

CP: is the primary author and was involved in the conception, design, data collection and drafting of the manuscript. MB: commenting and revision of the manuscript. IH: commenting and revision of the manuscript. KS: editing, revision and commenting on manuscript and ideas therein. DC: was involved in the conception, design, drafting and revision of the manuscript. All authors read and approved the final manuscript.

Acknowledgements

Funding support for the primary author by the BBSRC and the Kennel Club Charitable Trust (funding the Dogslife project) is gratefully acknowledged. The Roslin Institute is funded by an Institute Strategic Programme Grant from the BBSRC.

Received: 3 January 2014 Accepted: 25 March 2014 Published: 15 May 2014

References

- Dohoo I, Martin W, Stryhn H: Introduction to Observational Studies. In Vet Epidemiol Res. 2nd edition. Charlottetown, Canada: VER Inc; 2010:151–166.
- Thrusfield M: Observational Studies. In Vet Epidemiol. 3rd edition. Oxford: Blackwell Science Ltd; 2005;266–288.
- Mahmood SS, Levy D, Vasan RS, Wang TJ: The Framingham Heart Study and the epidemiology of cardiovascular disease: a historical perspective. Lancet 2013, 383:999–1008.
- 4. Doll R: Tobacco: a medical history. J Urban Health 1999, 76:289–313.
- Doll R, Peto R, Boreham J, Sutherland I: Mortality in relation to smoking: 50 years' observations on male British doctors. Br Med J 2004, 328:1519.
- Golding J: Children of the nineties. A longitudinal study of pregnancy and childhood based on the population of Avon. West Engl Med J 1990, 105:80–82.
- Richiardi L, Baussano I, Vizzini L, Douwes J, Pearce N, Merletti F: Feasibility
 of recruiting a birth cohort through the Internet: the experience of the
 NINFEA cohort. Eur J Epidemiol 2007, 22:831–837.
- Passaro K, Little R, Savitz D, Noss J: The effect of maternal drinking before conception and in early pregnancy on infant birthweight. *Epidemiology* 1996, 7:377–383.

- Overy C, Reynolds L, Tansey E: History of Avon Longitudinal Study of Parents and Children (ALSPAC), c.1980-2000. Wellcome Witnesses to Twentieth Century Medicine, Vol. 44. London: Queen Mary University of London; 2012:3–24.
- Golding J, Fleming P, Parkes J: Cot deaths and sleep position campaigns. Lancet 1992. 339:748–749.
- Hunt L, Fleming P, Golding J: Does the supine sleeping position have Any adverse effects on the child?: I. Health in the first six months. Pediatrics 1997. 100:e11–e11.
- Daniels JL, Longnecker MP, Rowland AS, Golding J: Fish intake during pregnancy and early cognitive development of offspring. Epidemiology 2004, 15:394–402.
- The Million Women Collaborative Study Group: The million women study: design and characteristics of the study population. Breast Cancer Res 1999, 1:73–80.
- Banks E, Beral V, Reeves G, Balkwill A, Barnes I: Fracture incidence in relation to the pattern of use of hormone therapy in postmenopausal women. J Am Med Assoc 2004, 291:2212–2220.
- The Million Women Collaborative Study Group: Breast cancer and hormone-replacement therapy: the million women study. Lancet 2003, 362:419–427.
- Deary IJ, Gow AJ, Pattie A, Starr JM: Cohort profile: the Lothian Birth Cohorts of 1921 and 1936. Int J Epidemiol 2012, 41:1576–1584.
- Schiepers OJG, Harris SE, Gow AJ, Pattie A, Brett CE, Starr JM, Deary IJ: APOE E4 status predicts age-related cognitive decline in the ninth decade: longitudinal follow-up of the Lothian Birth Cohort 1921. Mol Psychiatry 2012, 17:315–324.
- Gray G, Chesbrough K, Ryan M, Amoroso P, Boyko E, Gackstetter G, Hooper T, Riddle J: The millennium cohort study: a 21-year prospective cohort study of 140,000 military personnel. Mil Med 2002, 167:483–488.
- Smith B, Smith TC, Gray GC, Ryan MAK: When epidemiology meets the internet: Web-based surveys in the millennium cohort study. Am J Epidemiol 2007, 166:1345–1354.
- 20. NINFEA. [https://www.progettoninfea.it/index_en]
- 21. Dohoo I, Martin W, Stryhn H: Cohort Studies. In Vet Epidemiol Res. 2nd edition. Charlottetown, Canada: VER Inc; 2010:167–180.
- 22. O'Neill D, Church D, McGreevy P, Thomson P, Brodbelt D: Approaches to Canine health surveillance. *Genet Epidemiol Canine* 2014,1:2.
- Bonnett BN, Egenvall A: Age patterns of disease and death in insured Swedish dogs, cats and horses. J Comp Pathol 2010, 142:S33–S38.
- Egenvall A, Bonnett BN, Olson P, Hedhammar A: Validation of computerized Swedish dog and cat insurance data against veterinary practice records. Prev Vet Med 1998, 36:51–65.
- Egenvall A, Bonnett BN, Ohagen P, Olson P, Hedhammar A, von Euler H: Incidence of and survival after mammary tumors in a population of over 80,000 insured female dogs in Sweden from 1995 to 2002. Prev Vet Med 2005. 69:109–127.
- Bartlett PC, Van Buren JW, Neterer M, Zhou C: Disease surveillance and referral bias in the veterinary medical database. Prev Vet Med 2010, 94:264–271.
- Nødtvedt A, Guitian J, Egenvall A, Emanuelson U, Pfeiffer DU: The spatial distribution of atopic dermatitis cases in a population of insured Swedish dogs. Prev Vet Med 2007, 78:210–222.
- Glickman LT, Glickman NW, Moore GE, Lund EM, Lantz GC, Pressler BM: Association between chronic azotemic kidney disease and the severity of periodontal disease in dogs. Prev Vet Med 2011, 99:193–200.
- Courtenay O, Macdonald D, Lainson R, Shaw J, Dye C: Epidemiology of canine leishmaniasis: a comparative serological study of dogs and foxes in Amazon Brazil. Parasitology 1994, 109:273–279.
- Moreira ED, de Souza VM, Sreenivasan M, Lopes NL, Barreto RB, de Carvalho LP: Peridomestic risk factors for canine leishmaniasis in urban dwellings: new findings from a prospective study in Brazil. Am J Trop Med Hyg 2003, 60:203-207.
- Paranhos-Silva M, Nascimento E, Melro M, Oliveira G, dos Santos W, Pontesde-Carvalho L, Oliveira-dos-Santos A: Cohort study on canine emigration and Leishmania infection in an endemic area for American visceral leishmaniasis. Implications for the disease control. Acta Trop 1998, 69:75–83.
- Gramiccia M, Di Muccio T, Fiorentino E, Scalone A, Bongiorno G, Cappiello S, Paparcone R, Foglia Manzillo V, Maroli M, Gradoni L, Oliva G: Longitudinal study on the detection of canine Leishmania infections by conjunctival swab analysis and correlation with entomological parameters. Vet Parasitol 2010, 171:223–228.

- Oliva G, Scalone A, Foglia Manzillo V, Gramiccia M, Pagano A, Di Muccio T, Gradoni L: Incidence and time course of leishmania infantum infections examined by parasitological, serologic, and nested-PCR techniques in a cohort of Naive Dogs exposed to three consecutive transmission seasons. J Clin Microbiol 2006. 44:1318–1322.
- Otranto D, Testini G, Dantas-Torres F, Latrofa MS, Diniz de P PPV, de Caprariis D, Lia RP, Mencke N, Stanneck D, Capelli G, Breitschwerdt EB: Diagnosis of canine vector-borne diseases in young dogs: a longitudinal study. J Clin Microbiol 2010, 48:3316–3324.
- Quinnell R, Courtenay O, Garcez L, Kaye P, Shaw M, Dye C, Day M: IgG subclass responses in a longitudinal study of canine visceral leishmaniasis. Vet Immunol Immunopathol 2003, 91:161–168.
- Quinnell RJ, Carson C, Reithinger R, Garcez LM, Courtenay O: Evaluation of rK39 rapid diagnostic tests for canine visceral leishmaniasis: longitudinal study and meta-analysis. PLoS Negl Trop Dis 2013, 7:e1992.
- Moreira ED, Mendes de Souza VM, Sreenivasan M, Nascimento EG, Pontes de Carvalho L: Assessment of an optimized dog-culling program in the dynamics of canine Leishmania transmission. Vet Parasitol 2004, 122:245–252
- Kealy R, Lawler D, Ballam J, Mantz S, Biery D, Greeley E, Lust G, Segre M, Smith G, Stowe HD: Effects of diet restriction on life span and age-related changes in dogs. J Am Vet Med Assoc 2002, 220:1315–1320.
- Lawler DF, Evans RH, Larson BT, Spitznagel EL, Ellersieck MR, Kealy RD: Influence of lifetime food restriction on causes, time, and predictors of death in dogs. J Am Vet Med Assoc 2005, 226:225–231.
- Greeley EH, Spitznagel E, Lawler DF, Kealy RD, Segre M: Modulation of canine immunosenescence by life-long caloric restriction. Vet Immunol Immunopathol 2006, 111:287–299.
- Huck JL, Biery DN, Lawler DF, Gregor TP, Runge JJ, Evans RH, Kealy RD, Smith GK: A longitudinal study of the influence of lifetime food restriction on development of osteoarthritis in the canine elbow. Vet Surg 2009, 38:192–198.
- Kealy R, Olsson S, Monti K, Lawler D, Biery D, Helms R, Lust G, Smith G: Effects of limited food consumption on the incidence of hip dysplasia in growing dogs. J Am Vet Med Assoc 1992, 201:857–863.
- Kealy R, Lawler D, Ballam J, Lust G, Smith G, Biery D, Olsson S: Five-year longitudinal study on limited food consumption and development of osteoarthritis in coxofemoral joints of dogs. J Am Vet Med Assoc 1997, 210:222–225.
- Kealy R, Lawler D, Ballam J, Lust G, Biery D, Smith G, Mantz S: Evaluation of the effect of limited food consumption on radiographic evidence of osteoarthritis in dogs. J Am Vet Med Assoc 2000, 217:1678–1680.
- Powers M, Biery D, Lawler D, Evans R, Shofer F, Mayhew P, Gregor T, Kealy R, Smith G: Use of the caudolateral curvilinear osteophyte as an early marker for future development of osteoarthritis associated with hip dysplasia in dogs. J Am Vet Med Assoc 2004, 225:233–237.
- Runge JJ, Biery DN, Lawler DF, Gregor TP, Evans RH, Kealy RD, Szabo SD, Smith GK: The effects of lifetime food restriction on the development of osteoarthritis in the canine shoulder. Vet Surg 2008, 37:102–107.
- Smith GK, Paster ER, Powers MY, Lawler DF, Biery DN, Shofer FS, McKelvie PJ, Kealy RD: Lifelong diet restriction and radiographic evidence of osteoarthritis of the hip joint in dogs. J Am Vet Med Assoc 2006, 229:690–693.
- Smith GK, Lawler DF, Biery DN, Powers MY, Shofer F, Gregor TP, Karbe GT, McDonald-Lynch MB, Evans RH, Kealy RD: Chronology of hip dysplasia development in a cohort of 48 Labrador retrievers followed for life. Vet Surg 2012, 41:20–33.
- Szabo SD, Biery DN, Lawler DF, Shofer FS, Powers MY, Kealy RD, Smith GK: Evaluation of a circumferential femoral head osteophyte as an early indicator of osteoarthritis characteristic of canine hip dysplasia in dogs. J Am Vet Med Assoc 2007, 231:889–892.
- Dobson J, Hoather T, McKinley T, Wood J: Mortality in a cohort of flatcoated retrievers in the UK. Vet Comp Oncol 2009, 7:115–121.
- Thrusfield MV, Holt PE, Muirhead RH: Acquired urinary incontinence in bitches: its incidence and relationship to neutering practices. J Small Anim Pract 1998, 39:559–566.
- Nielen A, van der Gaag I, Knol B, Schukken Y: Investigation of mortality and pathological changes in a 14-month birth cohort of boxer puppies. Vet Rec 1998, 142:602–606.

- Van der Beek S, Nielen A, Schukken Y, Brascamp E: Evaluation of genetic, common-litter, and within-litter effects on preweaning mortality in a birth cohort of puppies. Am J Vet Res 1999, 60:1106–1110.
- Van Hagen M, Ducro B, van den Broek J, Knol B: Life expectancy in a birth cohort of Boxers followed up from weaning to 10 years of age. Am J Vet Res 2005, 66:1646–1650.
- Nielen A, Janss L, Knol B: Heritability estimations for diseases, coat color, body weight, and height in a birth cohort of Boxers. Am J Vet Res 2001, 62:1198–1206.
- Nielen A, Knol B, van Hagen M, van der Gaag I: Genetic and epidemiological investigation of a birth cohort of boxers. Tijdschr Diergeneeskd 2003, 138-596, 500.
- Van Hagen M, Ducro B, van den Broek J, Knol B: Incidence, risk factors, and heritability estimates of hind limb lameness caused by hip dysplasia in a birth cohort of Boxers. Am J Vet Res 2005, 66:307–312.
- Indrebø A, Trangerud C, Moe L: Canine neonatal mortality in four large breeds. Acta Vet Scand 2007, 49(Suppl 1):S2.
- Krontveit RI, Nødtvedt A, Sævik BK, Ropstad E, Skogmo HK, Trangerud C:
 A prospective study on canine hip dysplasia and growth in a cohort of four large breeds in Norway (1998–2001). Prev Vet Med 2010, 97:252–263.
- Sævik BK, Skancke EM, Trangerud C: A longitudinal study on diarrhoea and vomiting in young dogs of four large breeds. Acta Vet Scand 2012, 54:8–16
- 61. Dogslife. [http://www.dogslife.ac.uk]
- Clements DN, Handel IG, Rose E, Querry D, Pugh CA, Ollier WE, Morgan KL, Kennedy LJ, Sampson J, Summers KM, de Bronsvoort BMC: Dogslife: a webbased longitudinal study of Labrador Retriever health in the UK. BMC Vet Bes 2013. 9:13.
- Bonnett B, Egenvall A, Olson P, Hedhammar A: Mortality in insured Swedish dogs: rates and causes of death in various breeds. Vet Rec 1997, 141:40–44
- Kirkness EF, Bafna V, Halpern AL, Levy S, Remington K, Rusch DB, Delcher AL, Pop M, Wang W, Fraser CM, Venter JC: The dog genome: survey sequencing and comparative analysis. Science 2003, 301:1898–1903.
- 65. Lindblad-Toh K, Wade CM, Mikkelsen TS, Karlsson EK, Jaffe DB, Kamal M, Clamp M, Chang JL, Kulbokas EJ, Zody MC, Mauceli E, Xie X, Breen M, Wayne RK, Ostrander E, Ponting CP, Galibert F, Smith DR, DeJong PJ, Kirkness E, Alvarez P, Biagi T, Brockman W, Butler J, Chin C-W, Cook A, Cuff J, Daly MJ, DeCaprio D, Gnerre S, et al: Genome sequence, comparative analysis and haplotype structure of the domestic dog. Nature 2005, 438:803–819.

doi:10.1186/2052-6687-1-5

Cite this article as: Pugh et al.: What can cohort studies in the dog tell us?. Canine Genetics and Epidemiology 2014 1:5.

Submit your next manuscript to BioMed Central and take full advantage of:

- Convenient online submission
- Thorough peer review
- No space constraints or color figure charges
- Immediate publication on acceptance
- Inclusion in PubMed, CAS, Scopus and Google Scholar
- Research which is freely available for redistribution

Submit your manuscript at www.biomedcentral.com/submit

